# **Response to Reviewer Comments**

Dear [Reviewer's Name],

Thank you for taking the time to review our manuscript titled "[Manuscript Title]." We greatly appreciate your valuable feedback and suggestions. Below, we provide responses to each of your comments:

# **Reviewer Comment 1:**

[Insert reviewer comment here]

#### **Response:**

[Insert response here]

### **Reviewer Comment 2:**

[Insert reviewer comment here]

#### **Response:**

[Insert response here]

# **Reviewer Comment 3:**

[Insert reviewer comment here]

#### **Response:**

[Insert response here]

We have made the necessary revisions as indicated in the responses above and have attached the revised manuscript for your consideration. Thank you once again for your helpful comments.

Sincerely,
[Your Name]
[Your Institution]
[Your Contact Information]